OODS Tutorial 5

© 1999/2000 Michael Golm Universität Erlangen Nürnberg, Institut für Informatik 4

Homework Assignment #4: Drawing Editor GUI 0 points

Duringthis assignment you willimprovethe DrawingEditoryou programmed in the previous assignment. Copy all filesof assignment #3 in the assignment4 directory.Youcan also use the example solution that is available in/proj/i4oods/pub/assignment3.

a) The DrawingEditor shouldno longerrunas an applet -removethe init methodand make

DrawingEditor a subclass of Frame. Definea new class DrawingArea that should be a

subclass of Canvas. Add the DrawingArea to the DrawingEditor-Frame. Remove the

paint method fromthe DrawingEditor. Notethat younowhave to invokerepaint at the
DrawingArea. Remove mouseDown and mouseDrag fromthe DrawingEditor and use

the Java 1.1 event model to react to mouse events in the DrawingArea. b) Implement a GUI that allows the user to add shapes, and to save and load the drawing.

Use a FileDialog to let the user specifya filename. Lookfora descriptionofthe File-

Dialog class inthe "Java Core APIdocumentation".Toloadthe data fromthe fileyoushould

use the load method, that you implementedin the previous assignment. Implementa save method that saves the state of a drawing and call this method when the user presses the "Save" button. Toadd a shape, the user must type a shape name (class name), a position,and a dimension (likeRect 120 80 20 30) in the Shape-Input field and press the "Add" button. The layout of the user interface should be similar to the following figure: Add Save

Load

Drawing Area

Shape-Input

DrawingEditor